Author’s response to reviews

Title: Determinants of Traditional Medicine Utilization for Children: A Parental Level Study in Tole District, Oromia, Ethiopia

Authors:

Fekensa Hailu (hailufake@gmail.com)

Amsale Cherie (amsalec2002@yahoo.com)

Tigistu Gebreyohannis (tigistu.gebreyohannis@aau.edu.et)

Reta Hailu Belda (reta.hailu@aau.edu.et)

Version: 3 Date: 04 Apr 2020

Author’s response to reviews:

Responses to the second reviewers’ comments

Date: 04 April 2020

BMC Complementary & Alternative Medicine - Editorial Office
jeanelle.depadua@springernature.com
From: Reta Hailu Belda; reta.hailu@aau.edu.et
Subject: Response to the third reviewers’ comments to BMC Complementary Medicine and Therapies - BCAM-D-19-01836

Dear Editor, In-Hyuk Ha,

Thank you once again for your swift responses on the second revised version of our manuscript (Fekensa Hailu, Amsale Cherie, Tigistu Gebreyohannis, Reta Hailu Belda) entitled ‘Determinants of Traditional Medicine Utilization for Children: A Parental Level Study in Tole District, Oromia, Ethiopia (BCAM-D-19-01836)”. The authors have fully accepted your comments and revised the manuscript as follow.

Conclusion part of the Abstract, Page 1 L26-32 is summarized as follow : “Traditional medicine use is highly prevalent that its contribution to the public health is significant as some could not access to modern health service in the area. Therefore, health professionals should advise parents side-by-side procuring modern health services. In light of this, there is a need for further research on the efficacy and safety of traditional medicine for wider application.”

The main Conclusion P.20 Line 2, “…. There was high parental traditional medicine practice for children in this study (85.9%)”, “(85.9%)” is taken out as it was also explained in the result part.

We are excited to hear your decision soon!
Best regards,

Reta Hailu Belda (PhD)